Author's response to reviews

Title: Evaluation of a learner-designed course for teaching health research skills in Ghana

Authors:

Imelda Bates (ibates@liv.ac.uk)  
Daniel Ansong (ansongd@yahoo.com)  
George Bedu-Addo (gba@africaonline.com.gh)  
Tsiri Agbenyega (tsiri@ghana.com)  
Alex Yaw Osei Akoto (ale xo sei akoto@yahoo.com)  
Anthony Nsiah-Asare (anasare@africaonline.com.gh)  
Patrick Karikari (yawkarikari@hotmail.com)

Version: 5  Date: 26 June 2007

Author's response to reviews:

I have made all the changes requested in your e-mail (copied below)

Please indicate the corresponding author by means of an asterisk, or other superscript symbol, linking the name of the author to the words 'Corresponding author' and their address details. An email address is sufficient.

Please remove the repetition of the title of the manuscript from the page containing the Abstract. Please also remove the manuscript title from the Abstract section of the submission system.

Please rename Participants and methods --> Methods. You should include 'Participants' as the first subheading of your Methods section instead.

Please reformat your Conclusions section so that there are no bullet points; only continuous prose should be used.

Tables - Please note that we are unable to display vertical rules within the tables - please re-layout your table without these vertical elements.